UCSF UC San Francisco Previously Published Works

Title

Incident Heart Failure in Atherosclerotic Renal Artery Stenosis: A Post Hoc Analysis of the CORAL Trial.

Permalink https://escholarship.org/uc/item/0cg068px

Journal Kidney Medicine, 7(2)

Authors

Gupta, Rajesh Estrella, Michelle Scherzer, Rebecca <u>et al.</u>

Publication Date 2025-02-01

DOI

10.1016/j.xkme.2024.100948

Peer reviewed

Incident Heart Failure in Atherosclerotic Renal Artery Stenosis: A Post Hoc Analysis of the CORAL Trial

Rajesh Gupta, Michelle M. Estrella, Rebecca Scherzer, Pamela S. Brewster, Lance D. Dworkin, Hanh T. Nguyen, Yanmei Xie, Joachim H. Ix, Michael G. Shlipak, Timothy P. Murphy, Donald E. Cutlip, Eldrin F. Lewis, and Christopher J. Cooper

Rationale & Objective: Although renal artery stenosis (RAS) and heart failure (HF) have been linked, the incidence and predictors of HF among patients with RAS are not well described.

Study Design: Post hoc analysis of the Cardiovascular Outcomes in Renal Atherosclerotic Lesions (CORAL) multicenter, open-label, randomized controlled trial (RCT).

Settings and Participants: Patients with atherosclerotic RAS and elevated blood pressure, chronic kidney disease, or both, and without a history of HF at enrollment.

Intervention: Medical therapy alone versus medical therapy plus renal artery stenting.

Outcomes: Incident HF events.

Results: This analysis included 808 participants enrolled in the CORAL trial without evidence of baseline HF. During a median follow-up of 4.8 years, 54 participants (6.7%) developed incident HF. HF incidence rates did not differ by randomized

Renal artery stenosis (RAS) commonly affects patients with other manifestations of atherosclerosis, such as coronary artery disease or peripheral vascular disease (PVD),^{1,2} and has been associated with high rates of morbidity and mortality.³ RAS has been linked to heart failure (HF), with Pickering et al⁴ first describing the association in 1988. Since that time, several randomized controlled trials have investigated the role of RAS and renal artery interventional procedures in the management of hypertension and preservation of kidney function.⁵⁻⁷ However, the interrelationship between RAS and HF has received less attention.

Observational cohort studies of patients with HF have established that RAS and HF frequently coexist, with an estimated RAS prevalence of 15%-50% in patients with HF.⁸⁻

¹⁰ Furthermore, RAS is associated with increased rates of adverse events among patients with HF, such as worsening kidney function, recurrent HF hospitalizations, and death.¹⁰⁻ ¹³ Nonetheless, the diagnostic and management strategies for

preventing HF in patients with RAS remain uncertain.

The Cardiovascular Outcomes in Renal Atherosclerotic Lesions (CORAL) trial was the largest randomized controlled trial of persons with RAS. We sought to determine the incidence of HF among participants in CORAL and the effect of randomization to optimal

intervention (HR, 0.84; 95% confidence interval [CI], 0.49-1.43 for stent arm with medical arm as reference). Baseline diabetes (subdistribution hazard ratio (sHR), 2.07; 95% CI, 1.20-3.58), albuminuria (sHR, 1.12 per doubling of urinary albumin-creatinine ratio, 95% CI, 1.02-1.24), lower eGFR (sHR, 0.78 per 10 mL/min/1.73 m² estimated glomerular filtration rate calculated with cystatin C and creatinine, 95% CI, 0.69-0.88), and peripheral vascular disease (PVD) (sHR, 2.18, 95% CI, 1.21-3.91) were independent predictors of incident HF. Participants who experienced incident HF had greater kidney function decline before HF events.

Limitations: This is a post hoc analysis of a RCT. The number of HF events is small.

Conclusions: In patients with RAS, rates of incident HF did not differ between participants randomized to optimal medical therapy alone versus optimal medical therapy plus renal artery stenting. The presence of diabetes, PVD, and worse kidney health at baseline were associated with future HF events.



Correspondence to R. Gupta (rajesh.gupta@ utoledo.edu)

Kidney Med. 7(2):100948. Published online December 17, 2024.

doi: 10.1016/ j.xkme.2024.100948

© 2024 The Authors. Published by Elsevier Inc. on behalf of the National Kidney Foundation, Inc. This is an open access article under the CC BY license (http://creativecommons. org/licenses/by/4.0/).

medical therapy versus optimal medical therapy plus renal artery stent on the rate of incident HF events. Furthermore, we sought to identify baseline predictors of incident HF events.

METHODS

Study Design and Population

This is a post hoc cohort analysis of data from the CORAL trial (Clinical trial registration: NCT00081731) to assess the incidence of HF, the effect of the trial intervention on HF incidence, and predictors of HF among people with RAS. The CORAL trial was a prospective, international, multicenter clinical trial that compared the effects of optimal medical therapy alone or optimal medical therapy plus renal artery stenting on major cardiovascular or kidney events, defined as a composite of death from cardiovascular or kidney causes, stroke, myocardial infarction, hospitalization for congestive HF, progressive renal insufficiency, or the need for permanent kidney replacement therapy.⁷ Optimal medical therapy comprised treating blood pressure to goal of <140/90 mm Hg in general and to goal of <130/80 mm Hg among participants with diabetes or chronic kidney disease (CKD). Diabetes was treated to a hemoglobin A1c (HbA1c) goal of <7.0%,

Kidney Medicine



PLAIN-LANGUAGE SUMMARY

Renal artery stenosis has been linked with heart failure. The CORAL randomized controlled trial has the largest study population of participants with renal artery stenosis. In this analysis, we assessed the incidence and predictors of heart failure in CORAL. We found similar rates of incident heart failure among participants randomized to medical therapy alone vs. medical therapy plus renal artery stent. We identified independent predictors of incident heart failure among people with renal artery stenosis include PAD, diabetes, albuminuria, and lower baseline eGFR. In addition, eGFR declined prior to heart failure events.

based on national clinical practice guidelines current at the time of study. Comprehensive medical therapy included use of an angiotensin receptor-blocking drug (candesartan), a calcium channel blocker (amlodipine), a statin (atorvastatin), and antiplatelet therapy (aspirin). Thiazide diuretic was added if blood pressure goal was not met, and further protocol driven blood pressure control was implemented.

A total of 931 participants were assigned in a 1:1 ratio to either medical therapy alone or medical therapy plus renal artery stent. The inclusion criteria required an objectively measured RAS of at least 60%. All angiograms were centrally analyzed by the angiography core laboratory for the study at the University of Virginia. Trial enrollment began on May 16, 2005, and follow-up terminated September 28, 2012. Please see the original publication for details of the inclusion and exclusion criteria.⁷ For the present analysis, we excluded participants with baseline history of HF, resulting in a sample size of 808 participants.

All centers obtained institutional review board (IRB) approval and followed institutional policies and study protocol. The CORAL trial conduct adhered to the Declaration of Helsinki. All participating patients provided written informed consent. The present post hoc analysis of the CORAL clinical trial database was evaluated by the University of Toledo IRB. Because we used a deidentified database, this work is considered nonhuman subjects research and is exempt from IRB review. CORAL was registered with clinicaltrials.gov (NCT00081731).

Outcomes

The primary outcome for this analysis was incident HF. Hospitalization for congestive HF was an adjudicated trial endpoint, determined by a blinded clinical events committee, and it was defined as being hospitalized for 12 hours or longer because of documented signs and symptoms of HF and treatment with intravenous therapy (vasodilators, diuretics, or inotropes) during the hospital stay. Only the first HF event was counted to determine the number of incident HF events.

Candidate Covariates

All covariates were obtained at the baseline study visit and included age, gender, race, ethnicity, study arm, body mass index, systolic and diastolic blood pressure, urinary albumin-creatinine ratio (UACR), estimated glomerular filtration rate (eGFR), maximum renal artery percent stenosis, bilateral RAS, smoking within the past year, hyperlipidemia, history of myocardial infarction (MI), transient ischemic attack (TIA)/stroke, PVD, and diabetes mellitus. Hyperlipidemia was defined as total cholesterol >200 mg/ dL or treatment with a lipid lowering medication. Diabetes was defined as treatment with oral diabetes medication or insulin or fasting glucose > 126 mg/dL. Bilateral RAS was defined as ≥60% stenosis of bilateral renal arteries. Smoking status, PVD, TIA/stroke, and history of MI were patient reported. UACR was log-transformed to normalize its distribution. We used the CKD Epidemiology Collaboration (CKD EPI) equations to estimate eGFR with cystatin C, creatinine, and the combined equation.¹⁴

Statistical Analysis

We summarized demographic and baseline clinical characteristics stratified by intervention arm. We then compared incident HF rates by randomized intervention, baseline eGFR ≤ 60 versus ≥ 60 mL/min/1.73 m², and presence versus absence of baseline albuminuria. Because Kaplan-Meier curves can overestimate the incidence of the outcome over time, we instead plotted cumulative incidence accounting for the competing risk of death. We used Gray's test to compare differences in cumulative incidence by treatment arm.¹⁵ Multivariable Fine-Gray proportional subhazards regression models were used to assess risk factors for incident HF, evaluating the candidate covariates listed above while accounting for death as a competing risk.¹⁶ Proportional hazards assumptions were checked by comparing plots of log (-log(survival)) versus log of survival time and the Schoenfeld test. All candidate variables had <10% missingness except for UACR (11%) and stenosis (20%). Multiple imputation with the Markov chain Monte Carlo method for arbitrary missing multivariate normal data was used to impute missing covariates with 10 imputations.

We used Bayesian model averaging¹⁷ to identify parsimonious sets of risk factors that were independently associated with incident HF, retaining predictors with posterior probabilities >35%. This method accounts for model uncertainty in variable selection by averaging over all candidate models, each weighted by its posterior model probability. As a sensitivity analysis, we stratified the analysis by randomized intervention (Stent + Medical Therapy vs Medical Therapy arm) and tested interactions of each risk factor with the intervention.

Because development of HF may influence the trajectory of eGFR, we used joint models^{18,19} to simul-taneously evaluate repeated measures of eGFR and time-to-HF event data. This approach reduces bias and

improves precision over more traditional time-to-event analyses. Our joint model linked 2 sub-models: (1) a linear mixed effect model assessing changes in eGFR values across the baseline and follow-up study visits and (2) an exponential survival model beginning at the baseline study visit. Within the linear mixed effect submodel, we allowed the effect of time on eGFR trajectory to vary by HF status. This process was implemented by including a time-updated HF indicator variable for those who developed HF during the study, such that pre- and post-HF slopes could be estimated. For those who did not develop HF, a separate indicator variable was included to enable this group to have its own slope estimated separately.

Bayesian model averaging was performed using the Bayesian model averaging package from the R statistical computing language (R Development Core Team). All other analyses were conducted using the SAS system, version 9.4 (SAS Institute, Inc).

RESULTS

Among the 808 CORAL participants included in this analysis, 54 (6.7%) developed incident HF events during a median follow-up of 4.8 years. Baseline characteristics of participants are shown in Table 1; no differences were observed between the 2 trial arms among this subset of CORAL selected for absence of HF at baseline. There was no difference in the incidence of HF among participants randomized to optimal medical therapy versus renal artery stent plus optimal medical therapy (Fig 1).

In unadjusted analysis, older age, lower eGFR, higher UACR, TIA/stroke, PVD, and diabetes were associated with risk of incident HF. In a multivariable model assessing baseline variables, we identified diabetes, albuminuria, lower eGFR, and PVD as independent variables associated with incident HF (Table 2). We further assessed HF rates by eGFR and UACR categories and found the highest HF rates among participants with eGFR < 60 mL/min/ 1.73 m^2 and UACR > 30 mg/g (Fig 2). In stratified

Table	1. Baseline	Characteristics	in CORAL	After Excluding	Those With	Prevalent HF ^a

	% with Missing Values	Overall	Stent + Medical Therapy	Medical Therapy Only
Parameter		N = 808	N = 405	N = 403
Age, y	0	70 (64-77)	71 (64-76)	70 (63-77)
Female	0	413 (51%)	202 (50%)	211 (52%)
Race	0			
African American		55 (7%)	27 (7%)	28 (7%)
White		739 (91%)	372 (92%)	367 (91%)
Other		14 (2%)	6 (1%)	8 (2%)
Hispanic	0	45 (6%)	21 (5%)	24 (6%)
BMI (kg/m ²)	0.4%	28 (26-32)	28 (25-32)	28 (26-32)
Systolic BP (mm Hg)	0.9%	150 (135-165)	150 (135-165)	151 (134-166)
Diastolic BP (mm Hg)	0.9%	79 (71-88)	79 (71-87)	79 (71-88)
Creatinine (mg/dL)	5.1%	1.16 (0.90-1.48)	1.19 (0.91-1.49)	1.14 (0.89-1.46)
Cystatin C (mg/L)	5.0%	1.18 (0.95-1.46)	1.20 (0.96-1.47)	1.15 (0.94-1.45)
eGFRcyscr (race-free equation) (mL/min/1.73 m ²)	5.1%	60 (45-80)	60 (45-79)	60 (46-80)
UACR, mg/g	11%	21 (9-75)	22 (10-81)	19 (8-64)
Maximum renal artery percent stenosis	20%	68 (60-77)	69 (61-77)	67 (58-77)
Bilateral RAS	20%	146 (23%)	92 (24%)	54 (20%)
Smoking (within the past year)	0	246 (30%)	115 (28%)	131 (33%)
Hyperlipidemia	0.4%	707 (88%)	354 (88%)	353 (88%)
Myocardial infarction	0.1%	202 (25%)	95 (24%)	107 (27%)
TIA/stroke	2.1%	155 (20%)	80 (20%)	75 (19%)
Peripheral vascular disease	0.1%	391 (48%)	196 (49%)	195 (48%)
Diabetes mellitus	1.6%	246 (31%)	121 (30%)	125 (32%)
CKD ≥ Stage III	0%	502 (62%)	252 (62%)	250 (62%)
Diuretic use	1.1%			
Loop		102 (13%)	49 (12%)	53 (13%)
Thiazide		158 (20%)	71 (18%)	87 (22%)
Both		14 (2%)	5 (1%)	9 (2%)
None		525 (66%)	276 (69%)	249 (63%)

Abbreviations: BMI, body mass index; BP, blood pressure; CKD, chronic kidney disease; eGFR, estimated glomerular filtration rate by creatinine and cystatin C; RAS, renal artery stenosis; TIA, transient ischemic attack; UACR, urinary albumin-creatinine ratio.



Figure 1. Incidence of HF, stratified by randomization status. Cumulative incidence of HF accounting for competing risk of death. Comparison of differences in cumulative incidence of HF by randomized treatment arm using Gray's test.

analyses, the association of bilateral RAS with incident HF appeared to be modified by randomized intervention, with lower risk of incident HF in the renal artery stent plus medical therapy group (subdistribution hazard ratio [sHR], 0.53, 95% confidence interval [CI], 0.18-1.53) but higher risk in the medical therapy only group (sHR, 1.90; 95% CI, 0.73-4.93); however, the p-interaction of 0.07 did not reach statistical significance. The associations of the remaining baseline factors with incident HF did not appear to be influenced by assigned intervention group (Table 3).

Next, we assessed changes in eGFR during follow-up. Participants who experienced incident HF events had greater declines in eGFR compared with participants without HF events, and these declines occurred before HF events (Table 4).

In a sensitivity analysis, we assessed the entire CORAL cohort (N = 931, including participants with baseline history of HF). We found similar results overall. The rate of incident HF did not differ among participants

randomized to medical therapy versus medical therapy plus renal artery stent (Figure S1). Predictors of incident HF remained similar with estimated glomerular filtration rate by creatinine and cystatin C (eGFRcyscr), UACR, and diabetes being independent predictors. In the new model using the entire CORAL cohort, PVD was no longer significant, and age was a significant predictor of incident HF (Tables S1 and S2).

DISCUSSION

Among participants with RAS in the CORAL trial, the incidence of HF did not differ when patients were randomized to medical therapy plus renal artery stent versus medical therapy alone. Among persons with RAS, we found that factors independently associated with incident HF included diabetes, albuminuria, lower eGFR, and PVD, and participants who developed incident HF events had greater declines in eGFR before HF events than following the diagnosis of HF.

	Unadjusted	Fully adjusted ^b	Parsimonious
Parameter	SHR (95% CI)	SHR (95% CI)	SHR (95% CI)
Stent arm	0.84 (0.49-1.43), <i>P</i> = 0.52	0.67 (0.36-1.22), <i>P</i> = 0.19	0.73 (0.42-1.25), <i>P</i> = 0.25
Age (per decade)	1.43 (1.07-1.93), <i>P</i> = 0.02	1.24 (0.76-2.02), <i>P</i> = 0.38	
Male vs Female	1.45 (0.84-2.48), <i>P</i> = 0.18	1.76 (0.97-3.20), <i>P</i> = 0.064	
African American vs White	1.40 (0.56-3.49), <i>P</i> = 0.47	1.74 (0.65-4.63), <i>P</i> = 0.27	
Other vs White	1.11 (0.15-8.25), <i>P</i> = 0.92	1.31 (0.17-10.09), <i>P</i> = 0.80	
Hispanic	n/a – too sparse to estimate		
BMI (per kg/m²)	0.98 (0.94-1.03), <i>P</i> = 0.51	0.97 (0.91-1.03), <i>P</i> = 0.33	
Systolic BP (per 10 mm Hg)	1.09 (0.97-1.21), <i>P</i> = 0.15	1.05 (0.91-1.21), <i>P</i> = 0.52	
Diastolic BP (per 10 mm Hg)	0.89 (0.73-1.09), <i>P</i> = 0.27	0.85 (0.64-1.13), <i>P</i> = 0.27	
eGFRcyscr (per 10 mL/min/1.73 m²)	0.74 (0.66-0.83), <i>P</i> < 0.001	0.77 (0.67-0.89), <i>P</i> = 0.001	0.78 (0.69-0.89), <i>P</i> < 0.001
UACR (per doubling)	1.25 (1.14-1.37), <i>P</i> < 0.001	1.14 (1.02-1.28), <i>P</i> = 0.03	1.13 (1.02-1.25), <i>P</i> = 0.02
Max renal artery % stenosis	1.05 (0.80-1.39), <i>P</i> = 0.73	1.01 (0.74-1.37), <i>P</i> = 0.97	
Bilateral RAS	1.15 (0.59-2.22), <i>P</i> = 0.68	1.02 (0.48-2.17), <i>P</i> = 0.95	
Smoking (within past year)	0.83 (0.45-1.51), <i>P</i> = 0.53	1.05 (0.49-2.27), <i>P</i> = 0.89	
Hyperlipidemia	1.20 (0.47-3.05), <i>P</i> = 0.70	0.77 (0.29-2.09), <i>P</i> = 0.61	
Myocardial infarction	1.47 (0.83-2.60), <i>P</i> = 0.18	1.29 (0.70-2.41), <i>P</i> = 0.42	
TIA/stroke	1.89 (1.04-3.41), <i>P</i> = 0.04	1.42 (0.76-2.66), <i>P</i> = 0.28	
Peripheral vascular disease	2.32 (1.29-4.15), <i>P</i> = 0.005	1.93 (1.04-3.57), <i>P</i> = 0.04	2.23 (1.23-4.02), <i>P</i> = 0.008
Diabetes mellitus	2.41 (1.40-4.13), <i>P</i> = 0.001	2.19 (1.16-4.13), <i>P</i> = 0.02	2.03 (1.17-3.52), <i>P</i> = 0.01

Table 2. Associations of Baseline Characteristics With Incident HF in CORAL^a

Abbreviations: BMI, body mass index; BP, blood pressure; eGFR, estimated glomerular filtration rate by creatinine and cystatin C; RAS, renal artery stenosis; SHR, subdistribution hazard ratio; TIA, transient ischemic attack; UACR, urinary albumin-creatinine ratio.

^aHazard ratios with 95% confidence intervals from Fine-Gray competing risk models.

^bFully adjusted model controls for all risk factors simultaneously.

^cParsimonious model controls for reduced set of risk factors selected by Bayesian Model Averaging, with stent arm forced into the model.

To our knowledge, our study represents the largest report of the incidence and predictors of HF among persons with RAS. The relationship between RAS and HF has been known for many years, but high-quality data on this topic are limited. We found an approximately 7% (1.4% per year) incidence of HF in CORAL during study followup. This is higher than other contemporary clinical trials of individuals with hypertension, such as the SPRINT trial, which reported HF incidences of approximately 2.1% (0.64% per year).²⁰ Participants in CORAL had higher



Figure 2. Incident HF event rates by combined categories of baseline eGFRcyscr and UACR. $^{a}P < 0.05$ for comparison with control group (participants with ACR < 30 and eGFR ≥ 60).

Table 3. Associations Of Baseline Characteristics With Incident HF In CORAL, Stratified By Randomized Intervention Group^a

	Stent + Medical Rx	Medical Rx Only	Test for Interaction	
Parameter	SHR (95% CI)	SHR (95% CI)	P Value	
Age (per decade)	1.04 (0.55-1.97), <i>P</i> = 0.91	1.47 (0.85-2.54), <i>P</i> = 0.17	0.34	
Male vs Female	1.49 (0.64-3.48), <i>P</i> = 0.36	2.03 (0.91-4.53), <i>P</i> = 0.084	0.60	
African American vs White	0.83 (0.12-5.56), <i>P</i> = 0.8	2.40 (0.78-7.38), <i>P</i> = 0.13	0.34	
Other vs White	n/a – too sparse to estimate			
BMI (per kg/m²)	0.98 (0.91-1.05), <i>P</i> = 0.59	0.96 (0.88-1.05), <i>P</i> = 0.39	0.73	
Systolic BP (per 10 mm Hg)	1.09 (0.92-1.29), <i>P</i> = 0.31	1.01 (0.83-1.22), <i>P</i> = 0.93	0.47	
Diastolic BP (per 10 mm Hg)	0.97 (0.61-1.54), <i>P</i> = 0.90	0.78 (0.59-1.05), <i>P</i> = 0.10	0.37	
eGFRcyscr (per 10 mL/min/1.73 m ²)	0.77 (0.63-0.94), <i>P</i> = 0.009	0.78 (0.64-0.95), <i>P</i> = 0.01	0.92	
ACR (per doubling)	1.09 (0.93-1.27), <i>P</i> = 0.28	1.20 (1.03-1.41), <i>P</i> = 0.02	0.36	
Max renal artery % stenosis	1.02 (0.68-1.53), <i>P</i> = 0.93	1.00 (0.66-1.50), <i>P</i> = 0.99	0.94	
Bilateral RAS	0.53 (0.18-1.53), <i>P</i> = 0.24	1.90 (0.73-4.93), <i>P</i> = 0.19	0.07	
Smoking (within past year)	0.76 (0.24-2.35), <i>P</i> = 0.63	1.37 (0.55-3.39), <i>P</i> = 0.50	0.38	
Hyperlipidemia	0.43 (0.12-1.51), <i>P</i> = 0.19	1.30 (0.28-6.09), <i>P</i> = 0.74	0.27	
Myocardial infarction	0.95 (0.40-2.26), <i>P</i> = 0.90	1.70 (0.73-3.98), <i>P</i> = 0.22	0.34	
TIA/stroke	2.08 (0.93-4.68), <i>P</i> = 0.08	0.97 (0.36-2.56), <i>P</i> = 0.94	0.23	
Peripheral vascular disease	1.30 (0.55-3.10), <i>P</i> = 0.55	2.72 (1.14-6.47), <i>P</i> = 0.02	0.24	
Diabetes mellitus	2.10 (0.94-4.73), <i>P</i> = 0.07	2.27 (0.95-5.41), <i>P</i> = 0.07	0.90	

Abbreviations: BMI, body mass index; BP, blood pressure; eGFR, estimated glomerular filtration rate by creatinine and cystatin C; RAS, renal artery stenosis; SHR, subdistribution hazard ratio; TIA, transient ischemic attack; UACR, urinary albumin-creatinine ratio.

^aHazard ratios with 95% confidence intervals from Fine-Gray competing risk models. All estimates above are from fully adjusted models, controlling for all risk factors simultaneously.

baseline systolic blood pressure and lower eGFR compared with the participants enrolled in SPRINT, and SPRINT also excluded persons with diabetes. These factors may explain the higher HF incidence in CORAL.

We found a similar incidence of HF in CORAL regardless of whether participants were randomized to optimal medical therapy versus renal artery stent plus optimal medical therapy. These findings do not support renal artery stent treatment for prevention of HF. Additionally, these findings support the robust medical treatment in the CORAL trial which achieved excellent blood pressure control with high rates of treatment with angiotensin-converting enzyme inhibitors (ACE-I), angiotensin II receptor blockers (ARBs), calcium channel blockers, and thiazide diuretics. The presence of diabetes, albuminuria, lower eGFR, and PVD at baseline were independently associated with incident HF events. Diabetes and albuminuria reflect multifactorial kidney disease in participants with RAS, which may further predispose to HF events. Persons with RAS who have concomitant diabetes

Table 4. Estimated	Annual Changes in eGFR (mL/min/1.73 m^2)
Among Participants	With and Without Incident HF ^{a,b}

	Annual change in	
	eGFRcr (95%Cl)	P value
Slope in those with No HF	-1.5 (-1.7 to -1.2)	<i>P</i> < 0.001
Pre-event slope in those with HF	-3.6 (-5.1 to -2.2)	<i>P</i> < 0.001
Post-event slope in those with HF	-1.4 (-2.9 to +0.2)	<i>P</i> = 0.09

^aEstimates obtained from joint models of survival and longitudinal eGFR.

^bTests for differences: pre vs post HF: P = 0.003; no HF vs any HF: P = 0.01.

Kidney Med Vol 7 | Iss 2 | February 2025 | 100948

and mineralocorticoid receptor

and albuminuria may fair especially poorly. In a prior

CORAL manuscript, albuminuria was an independent

predictor of the composite CKD outcome, which was

defined as progression to end stage kidney disease, >30% decline in eGFR, or death caused by kidney disease.²¹ In

addition, participants in CORAL with greater than the

median levels of albuminuria (UACR > 22.5 mg/g) had higher risk of cardiovascular outcomes during study

follow-up.²² In the present analysis, the combination of

UACR \geq 30 mg/g and eGFR < 60 mL/min/1.73 m² was

associated with a substantially higher rate of incident HF.

In fact, there was a 7.6-fold increase in the rate of incident HF when comparing participants with UACR \ge 30 mg/g

and $eGFR < 60 \text{ mL/min/1.73 m}^2$ versus those with

UACR< 30 mg/g and eGFR \geq 60 mL/min/1.73 m² (Fig 2).

The presence of lower extremity PVD in this RAS cohort

represents a marker for higher total atherosclerosis burden.

Prior studies have found that polyvascular disease (atherosclerosis in more than 1 major vascular territory) is asso-

ciated with higher rates of HF and death compared with

Guidelines on RAS and HF give a class I recommendation for revascularization among patients with hemodynamically significant RAS and recurrent, unexplained HF or flash pulmonary edema but highlight the limited highquality data among patients with concomitant RAS and HF.^{26,27} Based on the findings from the current analysis, efforts should be made to optimize medical therapy for diabetes and albuminuria to reduce HF events in RAS populations. Future studies should investigate the safety and efficacy of renal protective medical therapies such as

atherosclerosis in a single vascular territory.²³⁻²⁵

SGLT2 inhibitors

antagonists for patients with RAS. A multi-targeted medical intervention of ACE-I/ARB, SGLT2 inhibitors, and mineralocorticoid receptor antagonists may potentially provide strong protection against HF events in patients with RAS.

Our study has several limitations that should be noted. This is a post hoc subgroup analysis of a randomized controlled trial. The number of participants that developed incident HF events is small. Echocardiographic data were not ascertained in the CORAL trial, so we cannot compare baseline ejection fraction. The bilateral RAS subgroup was small and the subgroup analysis for the effect of randomized treatment on incident HF may have been underpowered.

In conclusion, the rates of incident HF in the CORAL trial did not differ by randomization status. In patients with RAS, the presence of diabetes, PVD, albuminuria, and lower eGFR at baseline were each independently associated with incident HF. Evaluating trajectories of kidney function, eGFR tended to decline more before rather than after incident HF events in this population of hypertensive trial participants with RAS.

SUPPLEMENTARY MATERIALS

Supplementary File (PDF)

Figure S1: Incidence of HF, stratified by randomization status, including participants with and without prevalent HF.

Cumulative incidence of HF accounting for competing risk of death. Comparison of differences in cumulative incidence of HF by randomized treatment arm using Gray's test.

 Table S1: Baseline Characteristics in CORAL, Including Those With and Without Prevalent HF.

 Table S2:
 Predictors of Incident HF in a Multivariable Model, Including Those With and Without Prevalent HF.

ARTICLE INFORMATION

Authors' Full Names and Academic Degrees: Rajesh Gupta, MD, Michelle M. Estrella, MD, MHS, Rebecca Scherzer, PhD, Pamela S. Brewster, MA, Lance D. Dworkin, MD, Hanh T. Nguyen, PhD, Yanmei Xie, PhD, Joachim H. Ix, MD, Michael G. Shlipak, MD, MPH, Timothy P. Murphy, MD, Donald E. Cutlip, MD, Eldrin F. Lewis, MD, MPH, Christopher J. Cooper, MD

Authors' Affiliations: Division of Cardiovascular Medicine (RG, JCJ) and Department of Medicine (RG, PSB, LDD, HTN, YX), University of Toledo College of Medicine and Life Sciences, Toledo, OH; Kidney Health Research Collaborative (MME, RS, MGS), San Francisco Veterans Affairs Healthcare System and University of California, San Francisco, CA; Nephrology Section (JHI), Veterans Affairs San Diego Healthcare System and Division of Nephrology-Hypertension, University of California, San Diego, CA; Summa Therapeutics, LLC (TPM), Cambridge, MA; Beth Israel Deaconess Medical Center and Baim Institute for Clinical Research (DEC), Boston, MA; Cardiovascular Division (EFL), Stanford University, Stanford, CA.

Address for Correspondence: Rajesh Gupta, MD, University of Toledo, Division of Cardiovascular Medicine, 3000 Arlington Avenue, MS#1118, Toledo, Ohio 43614. Email: rajesh.gupta@utoledo.edu

Authors' Contributions: Each author contributed important intellectual content during manuscript drafting or revision and accepts accountability for the overall work by ensuring that questions pertaining to the accuracy or integrity of any portion of the work are appropriately investigated and resolved.

Support: CORAL was supported by grants (U01HL071556, U01HL072734, U01HL072735, U01HL072736, and U01HL072737) from the National Heart, Lung, and Blood Institute of the National Institutes of Health. Additional financial support was provided by Pfizer and Cordis. Study drugs were supplied by AstraZeneca (Atacand, Atacand-HCT) and Pfizer (Caduet). Study devices were from Cordis (Genesis stent).

Investigators RS, JHI, and MGS are supported by 5R01DK098234. Investigators RG, RS, JHI, MME, and MGS are supported by 1R01HL175306.

Financial Disclosure: Rajesh Gupta has received grant support for NIH 1R01HL175306. Michelle M. Estrella has served on advisory boards for Boehringer Ingelheim and AstraZeneca. She has a collaborative research agreement with Bayer, Inc. Grant support from 1R01HL175306. Rebecca Scherzer received grant support from NIH 5R01DK098234 and 1R01HL175306. Joachim H. Ix holds an investigator-initiated grant from the Juvenile Diabetes Research Foundation, and has served on advisory boards for Akebia, AstraZeneca, and Bayer. Grant support from NIH 5R01DK098234 and 1R01HL175306. Michael G. Shlipak has served on advisory boards for Boehringer Ingelheim, AstraZeneca, and Bayer, and received research support from Bayer and grant support from NIH 5R01DK098234 and 1R01HL175306. The remaining authors declare that they have no relevant financial interests.

Peer Review: Received June 12, 2024. Evaluated by 2 external peer reviewers, with direct editorial input from the Statistical Editor and the Editor-in-Chief. Accepted in revised form November 1, 2024.

REFERENCES

- Buller CE, Nogareda JG, Ramanathan K, et al. The profile of cardiac patients with renal artery stenosis. J Am Coll Cardiol. 2004;43:1606-1613.
- Boateng FK, Greco BA. Renal artery stenosis: prevalence of, risk factors for, and management of in-stent stenosis. *Am J Kidney Dis.* 2013;61:147-160.
- **3.** Tsao CW, Aday AW, Almarzooq ZI, et al. Heart disease and stroke statistics-2022 update: a report from the American Heart Association. *Circulation*. 2022;145:e153-e639.
- 4. Pickering TG, Herman L, Devereux RB, et al. Recurrent pulmonary oedema in hypertension due to bilateral renal artery stenosis: treatment by angioplasty or surgical revascularisation. *Lancet.* 1988;2:551-552.
- van Jaarsveld BC, Krijnen P, Pieterman H, et al. The effect of balloon angioplasty on hypertension in atherosclerotic renalartery stenosis. Dutch Renal Artery Stenosis Intervention Cooperative Study Group. N Engl J Med. 2000;342:1007-1014.
- Investigators A, Wheatley K, Ives N, et al. Revascularization versus medical therapy for renal-artery stenosis. *N Engl J Med.* 2009;361:1953-1962.
- Cooper CJ, Murphy TP, Cutlip DE, et al. Stenting and medical therapy for atherosclerotic renal-artery stenosis. *N Engl J Med.* 2014;370:13-22.
- 8. Emans ME, van der Putten K, Velthuis BK, et al. Atherosclerotic renal artery stenosis is prevalent in cardiorenal patients but not associated with left ventricular function and myocardial fibrosis as assessed by cardiac magnetic resonance imaging. *BMC Cardiovasc Disord.* 2012;12:76.
- MacDowall P, Kalra PA, O'Donoghue DJ, Waldek S, Mamtora H, Brown K. Risk of morbidity from renovascular

disease in elderly patients with congestive cardiac failure. *Lancet.* 1998;352:13-16.

- Bourantas CV, Loh HP, Lukaschuk EI, et al. Renal artery stenosis: an innocent bystander or an independent predictor of worse outcome in patients with chronic heart failure? A magnetic resonance imaging study. *Eur J Heart Fail.* 2012;14:764-772.
- Zheng B, Ma Q, Zheng LH, Yong Q, He YH, Liu JH. Analysis of renal artery stenosis in patients with heart failure: A RASHEF Study. *Chin Med J (Engl)*. 2015;128:2777-2782.
- de Silva R, Loh H, Rigby AS, et al. Epidemiology, associated factors, and prognostic outcomes of renal artery stenosis in chronic heart failure assessed by magnetic resonance angiography. *Am J Cardiol.* 2007;100:273-279.
- Devanaboyina M, Shastri P, Thompson N, et al. Impact of renal artery stenosis on acute kidney injury and outcomes after heart failure hospitalization. *Am J Med Sci.* 2023;365:242-248.
- Inker LA, Eneanya ND, Coresh J, et al. New creatinine- and cystatin C-based equations to estimate GFR without race. *N Engl J Med.* 2021;385:1737-1749.
- Gray RJ. A class of K-sample tests for comparing the cumulative incidence of a competing risk. Ann Stat. 1988;16:1141-1154.
- Fine JP, Gray RJ. A proportional hazards model for the subdistribution of a competing risk. J Am Stat Assoc. 1999;94: 496-509.
- 17. Hoeting JA, Madigan D, Raftery AE, Volinsky CT. Bayesian model averaging: A tutorial. *Stat Sci.* 1999;14:382-401.
- Garcia-Hernandez A, Rizopoulos D. %JM: A SAS macro to fit jointly generalized mixed models for longitudinal data and timeto-event responses. J Stat Softw. 2018;84:1-29.
- Wulfsohn MS, Tsiatis AA. A joint model for survival and longitudinal data measured with error. *Biometrics*. 1997;53:330-339.
- Group SR, Wright JT Jr, Williamson JD, et al. A randomized trial of intensive versus standard blood-pressure control. N Engl J Med. 2015;373:2103-2116.
- Tuttle KR, Dworkin LD, Henrich W, et al. Effects of stenting for atherosclerotic renal artery stenosis on eGFR and predictors of clinical events in the CORAL trial. *Clin J Am Soc Nephrol.* 2016;11:1180-1188.

- 22. Murphy TP, Cooper CJ, Pencina KM, et al. Relationship of albuminuria and renal artery stent outcomes: results from the CORAL randomized clinical trial (cardiovascular outcomes with renal artery lesions). *Hypertension*. 2016;68:1145-1152.
- 23. Khan MS, Anker SD, Filippatos G, et al. Vascular disease burden, outcomes and benefits with empagliflozin in heart failure: insights from the EMPEROR-reduced trial. *J Card Fail.* 2023;29:1345-1354.
- Freedman BL, Berg DD, Scirica BM, et al. Epidemiology of heart failure hospitalization in patients with stable atherothrombotic disease: Insights from the TRA 2 degrees P-TIMI 50 trial. *Clin Cardiol.* 2022;45:831-838.
- Verma S, Mazer CD, Inzucchi SE, et al. Impact of polyvascular disease with and without co-existent kidney dysfunction on cardiovascular outcomes in diabetes: A post hoc analysis of EMPA-REG OUTCOME. *Diabetes Obes Metab.* 2021;23:1173-1181.
- 26. Hirsch AT, Haskal ZJ, Hertzer NR, et al. ACC/AHA 2005 Practice Guidelines for the management of patients with peripheral arterial disease (lower extremity, renal, mesenteric, and abdominal aortic): a collaborative report from the American Association for Vascular Surgery/Society for Vascular Surgery, Society for Cardiovascular Angiography and Interventions, Society for Vascular Medicine and Biology, Society of Interventional Radiology, and the ACC/AHA Task Force on Practice Guidelines (Writing Committee to Develop Guidelines for the Management of Patients With Peripheral Arterial Disease): endorsed by the American Association of Cardiovascular and Pulmonary Rehabilitation; National Heart, Lung, and Blood Institute; Society for Vascular Nursing; TransAtlantic Inter-Society Consensus; and Vascular Disease Foundation. Circulation. 2006;113:e463-e654.
- 27. Anderson JL, Halperin JL, Albert NM, et al. Management of patients with peripheral artery disease (compilation of 2005 and 2011 ACCF/AHA guideline recommendations): a report of the American College of Cardiology Foundation/American Heart Association Task Force on Practice Guidelines. *Circulation.* 2013;127:1425-1443.